\begin{thebibliography}{10}
\providecommand{\url}[1]{\texttt{#1}}
\providecommand{\urlprefix}{\\URL: }
\providecommand{\bibAnnoteFile}[1]{%
  \IfFileExists{#1}{\begin{quotation}\noindent\textsc{Key:} #1\\
  \textsc{Annotation:}\ \input{#1}\end{quotation}}{}}
\providecommand{\bibAnnote}[2]{%
\begin{quotation}\noindent\textsc{Annotation:}\ #2\end{quotation}}

\bibitem{Geesaman:2015fha}
A.~Aprahamian \emph{et~al.}, \emph{{Reaching for the Horizon: The 2015 Long
  Range Plan for Nuclear Science}}, Nuclear Science Advisory Committee  (2015),
  \urlprefix\url{{http://science.energy.gov/~/media/np/nsac/pdf/2015LRP/2015_LRPNS_091815.pdf}}
\bibAnnoteFile{Geesaman:2015fha}

\bibitem{Olive:2016xmw}
C.~Patrignani, \emph{{Review of Particle Physics}}, Chin. Phys. \textbf{C40}
  (2016)~(10), p. 100001
\bibAnnoteFile{Olive:2016xmw}

\bibitem{Dugger:2007bt}
M.~Dugger \emph{et~al.}, \emph{{$\pi^0$ photoproduction on the proton for
  photon energies from 0.675 to 2.875-GeV}}, Phys. Rev. \textbf{C76} (2007), p.
  025211
\bibAnnoteFile{Dugger:2007bt}

\bibitem{Dugger:2013crn}
M.~Dugger \emph{et~al.}, \emph{{Beam asymmetry $\Sigma$ for $\pi^+$ and $\pi^0$
  photoproduction on the proton for photon energies from 1.102 to 1.862 GeV}},
  Phys. Rev. \textbf{C88} (2013)~(6), p. 065203, [Addendum: Phys.
  Rev.C89,no.2,029901(2014)]
\bibAnnoteFile{Dugger:2013crn}

\bibitem{Dugger:2009pn}
M.~Dugger \emph{et~al.}, \emph{{$\pi^+$ photoproduction on the proton for
  photon energies from 0.725 to 2.875-GeV}}, Phys. Rev. \textbf{C79} (2009), p.
  065206
\bibAnnoteFile{Dugger:2009pn}

\bibitem{Dugger:2002ft}
M.~Dugger \emph{et~al.}, \emph{{Eta photoproduction on the proton for photon
  energies from 0.75-GeV to 1.95-GeV}}, Phys. Rev. Lett. \textbf{89} (2002), p.
  222002, [Erratum: Phys. Rev. Lett.89,249904(2002)]
\bibAnnoteFile{Dugger:2002ft}

\bibitem{Collins:2017sgu}
P.~Collins \emph{et~al.}, \emph{{Photon beam asymmetry $\Sigma$ for $\eta$ and
  $\eta^\prime$ photoproduction from the proton}}, Phys. Lett. \textbf{B771}
  (2017), pp. 213--221
\bibAnnoteFile{Collins:2017sgu}

\bibitem{Senderovich:2015lek}
I.~Senderovich \emph{et~al.}, \emph{{First measurement of the helicity
  asymmetry $E$ in $\eta$ photoproduction on the proton}}, Phys. Lett.
  \textbf{B755} (2016), pp. 64--69
\bibAnnoteFile{Senderovich:2015lek}

\bibitem{Dugger:2005my}
M.~Dugger \emph{et~al.}, \emph{{$\eta$$\prime$ photoproduction on the proton
  for photon energies from 1.527-GeV to 2.227-GeV}}, Phys. Rev. Lett.
  \textbf{96} (2006), p. 062001, [Erratum: Phys. Rev. Lett.96,169905(2006)]
\bibAnnoteFile{Dugger:2005my}

\bibitem{Collins:2017vev}
P.~Collins \emph{et~al.}, \emph{{Photon beam asymmetry $\Sigma$ in the reaction
  $\vec{\gamma} p \to p \omega$ for $E_\gamma$ = 1.152 to 1.876 GeV}}, Phys.
  Lett. \textbf{B773} (2017), pp. 112--120
\bibAnnoteFile{Collins:2017vev}

\bibitem{CascProp}
A.~Afanasev \emph{et~al.}, \emph{{Photoproduction of the Very Strangest Baryons
  on a Proton Target in CLAS12}}, {Jefferson Lab Experiment Proposal
  E12-11-008}, CLAS Collaboration, Jefferson Lab (2012),
  \urlprefix\url{{http://www.jlab.org/exp_prog/proposals/12/PR12-12-008.pdf}}
\bibAnnoteFile{CascProp}

\bibitem{Capstick:1986bm}
S.~Capstick and N.~Isgur, \emph{{Baryons in a relativized quark model with
  chromodynamics}}, Phys. Rev. \textbf{D34} (1986), p. 2809, [AIP Conf.
  Proc.132,267(1985)]
\bibAnnoteFile{Capstick:1986bm}

\bibitem{Crede:2017kyh}
V.~Crede and S.~Capstick, \emph{{Bound States of $N$'s and $\Xi$'s}}, in
  \emph{{Workshop on Excited Hyperons in QCD Thermodynamics at Freeze-Out
  (YSTAR2016) Mini-Proceedings}} (2017), pp. 40--45,
  \urlprefix\url{http://inspirehep.net/record/1511541/files/1510585_40-45.pdf}
\bibAnnoteFile{Crede:2017kyh}

\bibitem{Adamovich:1997ud}
M.~I. Adamovich \emph{et~al.}, \emph{{First observation of the Xi- pi+ decay
  mode of the Xi(1690)0 hyperon}}, Eur. Phys. J. \textbf{C5} (1998), pp.
  621--624
\bibAnnoteFile{Adamovich:1997ud}

\bibitem{Hassall:1981fs}
J.~K. Hassall, R.~E. Ansorge, J.~R. Carter, W.~W. Neale, J.~G. Rushbrooke,
  D.~R. Ward, B.~Y. Oh, M.~Pratap, G.~A. Smith and J.~Whitmore,
  \emph{{Production of S = -2 and -3 Baryon States in 6.5-{GeV}/$c K^- P$
  Interactions}}, Nucl. Phys. \textbf{B189} (1981), pp. 397--420
\bibAnnoteFile{Hassall:1981fs}

\bibitem{Criegee:1969qg}
L.~Criegee \emph{et~al.}, \emph{{$\rho$ production with polarized photons}},
  Phys. Lett. \textbf{28B} (1968), pp. 282--286
\bibAnnoteFile{Criegee:1969qg}

\bibitem{Ballam:1971yd}
J.~Ballam \emph{et~al.}, \emph{{Bubble Chamber Study of Photoproduction by
  2.8-GeV and 4.7-GeV Polarized Photons. 1. Cross-Section Determinations and
  Production of rho0 and Delta++ in the Reaction gamma p --> p pi+ pi-}}, Phys.
  Rev. \textbf{D5} (1972), p. 545
\bibAnnoteFile{Ballam:1971yd}

\bibitem{Dudek:2013yja}
J.~J. Dudek, R.~G. Edwards, P.~Guo and C.~E. Thomas, \emph{{Toward the excited
  isoscalar meson spectrum from lattice QCD}}, Phys. Rev. \textbf{D88}
  (2013)~(9), p. 094505
\bibAnnoteFile{Dudek:2013yja}

\bibitem{Chung:2002fz}
S.~U. Chung, E.~Klempt and J.~G. Korner, \emph{{SU(3) classification of
  $p-$wave $\eta \pi$ and $\eta^\prime \pi$ systems}}, Eur. Phys. J.
  \textbf{A15} (2002), pp. 539--542
\bibAnnoteFile{Chung:2002fz}

\bibitem{Iddir:2000yb}
F.~Iddir and A.~S. Safir, \emph{{The Decay of the observed $J^{PC} =
  1^{-+}(1400)$ and $J^{PC}=1^{-+}(1600)$ hybrid candidates}}, Phys. Lett.
  \textbf{B507} (2001), pp. 183--192
\bibAnnoteFile{Iddir:2000yb}

\bibitem{Close:1987aw}
F.~E. Close and H.~J. Lipkin, \emph{{New Experimental Evidence for Four Quark
  Exotics:The Serpukhov $\phi \pi$ Resonance and the Gams $\eta \pi$
  Enhancement}}, Phys. Lett. \textbf{B196} (1987), p. 245
\bibAnnoteFile{Close:1987aw}

\bibitem{AlGhoul:2017nbp}
H.~Al~Ghoul \emph{et~al.}, \emph{{Measurement of the beam asymmetry $\Sigma$
  for $\pi^0$ and $\eta$ photoproduction on the proton at $E_\gamma = 9$ GeV}},
  Phys. Rev. \textbf{C95} (2017)~(4), p. 042201
\bibAnnoteFile{AlGhoul:2017nbp}

\bibitem{gluexPac30}
G.~Collaboration, \emph{{Mapping the Spectrum of Light Quark Mesons and Gluonic
  Excitations with Linearly Polarized Photons}}, {Jefferson Lab Experiment
  Proposal PR-06-102}, GlueX Collaboration, Jefferson Lab (2006),
  \urlprefix\url{{https://halldweb.jlab.org/DocDB/0012/001226/001/pac30.pdf}}
\bibAnnoteFile{gluexPac30}

\bibitem{gx2year}
C.~Meyers, \emph{{Potential GlueX Physics During the First Two Years of 12 GeV
  Running}}, {Report}, GlueX Collaboration, Jefferson Lab (2006),
  \urlprefix\url{{https://halldweb.jlab.org/DocDB/0006/000621/002/gluex_pac.pdf}}
\bibAnnoteFile{gx2year}

\bibitem{Adam:2004fq}
I.~Adam \emph{et~al.}, \emph{{The DIRC particle identification system for the
  BaBar experiment}}, Nucl. Instrum. Meth. \textbf{A538} (2005), pp. 281--357
\bibAnnoteFile{Adam:2004fq}

\bibitem{Dugger:2017zoq}
M.~Dugger \emph{et~al.}, \emph{{Design and construction of a high-energy photon
  polarimeter}}, Nucl. Instrum. Meth. \textbf{A867} (2017), pp. 115--127
\bibAnnoteFile{Dugger:2017zoq}

\bibitem{Anderson:1971xh}
R.~L. Anderson, D.~Gustavson, J.~R. Johnson, I.~Overman, D.~Ritson, B.~H. Wiik
  and D.~Worcester, \emph{{HIGH-ENERGY pi0 PHOTOPRODUCTION FROM HYDROGEN WITH
  UNPOLARIZED AND LINEARLY POLARIZED PHOTONS}}, Phys. Rev. \textbf{D4} (1971),
  p. 1937
\bibAnnoteFile{Anderson:1971xh}

\bibitem{Adams:2001sk}
G.~S. Adams \emph{et~al.}, \emph{{Observation of pseudoscalar and axial vector
  resonances in pi- p ---> K+ K- pi0 n at 18-GeV}}, Phys. Lett. \textbf{B516}
  (2001), pp. 264--272
\bibAnnoteFile{Adams:2001sk}

\bibitem{clasDatabase}
{CLAS Collaboration}, \emph{JLAB Experiment {CLAS} Database},
  \urlprefix\url{http://clasweb.jlab.org/physicsdb/intro.html}
\bibAnnoteFile{clasDatabase}

\bibitem{SAIDdatabase}
{Institute for Nuclear Studies Data Analysis Center}, \emph{{SAID Partial-Wave
  Analysis Facility}}, \urlprefix\url{http://gwdac.phys.gwu.edu/}
\bibAnnoteFile{SAIDdatabase}

\bibitem{evioUserGuide}
{Jefferson Lab Data Acquisition Group}, \emph{EVIO Users' Guide},
  \urlprefix\url{ftp://ftp.jlab.org/pub/coda/evio/2.0/evio_Users_Guide.pdf}
\bibAnnoteFile{evioUserGuide}

\end{thebibliography}

\section{Research to be undertaken under proposed award}

\label{sec:overview}                                    

The ASU Meson Physics Group played a central role
in designing, building, commissioning, and exploiting
the past and current generation of experimental facilities at JLab by
leading and collaborating in experiments in Hall B
that advanced understanding of the nucleon.
The ASU group subsequently led and participated in a program of experiments
that provided the bulk of the current database for non-strange meson photoproduction data,
which underpin the current understanding of the nucleon.
This program included the development and exploitation of polarized photon beams and polarized
proton targets. ASU-led experiments and analyses using CLAS are summarized in Table  \ref{tbl:mpg}.
(Remaining 6 GeV CLAS analyses will be completed by the start of
the requested funding period.)
\vspace{12pt}

\begin{table}[h]
\caption{\small{ASU-led research in the CLAS 6 GeV program. Citation statistics from INSPIRE (8/22/2018).
(The project marked with ``$\ast$'' has statistics insufficient for publication.)
}}
\vspace{2mm}
\centering
\begin{tabular}{|c|c|c|}
\hline
Reaction & \ Observable & \ Status [Presentations, publications]  \\
\hline
\hline
\multirow{2}{*}{$\gamma p \rightarrow p \pi^0$}
& $d\sigma/d\Omega$ & Completed. Published August 2007. \cite{Dugger:2007bt} (cited 87 times)  \\
\cline{2-3}
& $\Sigma$ &Completed. Published December 2013. \cite{Dugger:2013crn} (cited 32 times)\\
\hline
\multirow{2}{*}{$\gamma p \rightarrow n \pi^+$}
& $d\sigma/d\Omega$ & Completed.  Published June 2009. \cite{Dugger:2009pn} (cited 103 times)  \\
\cline{2-3}
& $\Sigma$ &Completed. Published December 2013. \cite{Dugger:2013crn} (cited 32 times)\\
\cline{2-3}
& $T, F$ &Analysis note being drafted. \\
\cline{2-3}
& $P, H$ &Analysis note being drafted.  \\
\hline
\multirow{2}{*}{$\gamma p \rightarrow p \eta$}
& $d\sigma/d\Omega$ & Completed. Published November 2002. \cite{Dugger:2002ft} (cited 129 times) \\
\cline{2-3}
& $\Sigma$ &Completed. Published March 2017. \cite{Collins:2017sgu} (cited 13 times). \\
\cline{2-3}
& $E$ &Completed. Published  April 2016. \cite{Senderovich:2015lek} (cited 23 times)\\
\cline{2-3}
& $T, F$ &Dissertation of Ross Tucker successfully defended April 2016. $\ast$\\
\hline
\multirow{2}{*}{$\gamma p \rightarrow p \eta^\prime$}
& $d\sigma/d\Omega$ & Completed. Published February 2006. \cite{Dugger:2005my} (cited 65 times) \\
\cline{2-3}
& $\Sigma$ &Completed. Published March 2017. \cite{Collins:2017sgu} (cited 13 times) \\
\hline
$\gamma p \rightarrow p \omega$ & $\Sigma$ &Completed. Published June 2017. \cite{Collins:2017vev} (cited 3 time\
s) \\
\hline
\end{tabular}
 \label{tbl:mpg}
\end{table}

This proposal capitalizes on the momentum and experience of the ASU group 
to pursue a vigorous experimental program with the newly-upgraded facilities at JLab.
We will use the CLAS12 detector to study multi-strange-quark baryons in Hall B, 
and will employ the GlueX detector in Hall D to search for hybrid mesons, while
also providing precise and reliable characterization of the Hall D photon beam for all Hall D
experiments using the ASU triplet polarimeter (TPOL). 


